Author's response to reviews

Title: Prevalence, trends, patterns and associations of analgesic use in Germany

Authors:

Giselle Sarganas Dr (sarganas-margolisg@rki.de)  
Amanda K Buttery Dr (a.buttery@nhs.net)  
Wanli Zhuang (zhuangw@rki.de)  
Ingrid-Katharina Wolf (wolfi@rki.de)  
Daniel Grams (grams@rki.de)  
Angelika Schaffrath Rosario Dr (schaffrath-rosarioa@rki.de)  
Christa Scheidt-Nave Dr (scheidt-navec@rki.de)  
Hildtraud Knopf Dr (knopfh@rki.de)

Version: 3  
Date: 7 September 2015

Author's response to reviews: see over
Reviewer’s Response Cover letter MS: 4944903181675786

Prevalence, trends, patterns and associations of analgesic use in Germany

Thank you very much. As your request, we have corrected the title page as followed:

- Moved CI from the title page, and used the correct section heading
- We have deleted sources of support from the title page -GC